Author's response to reviews

Title: Prescriber and staff perceptions of an electronic prescribing system in primary care: A qualitative assessment

Authors:

Emily Beth Devine (bdevine@uw.edu)
Emily C Williams (Emily.Williams3@va.gov)
Diane P Martin (dianemar@uw.edu)
Dean F Sittig (Dean.F.Sittig@uth.tmc.edu)
Peter Tarczy-Hornoch (pth@uw.ued)
Thomas H Payne (tpayne@uw.edu)
Sean D Sullivan (sdsull@uw.edu)

Version: 4 Date: 3 November 2010

Author's response to reviews: see over
November 2, 2010

Philippa Harris, PhD  
Executive Editor in training  
BMC-series Journals  
BiomMed Central  
Floor 6, 236 Gray’s Inn Road  
London, WC1X 8HL

Re: MS: 3108147684089155, for BMC Medical Informatics and Decision-Making

Dear Dr. Harris:

We are pleased to submit our finalized manuscript, “Prescriber and staff perceptions of an electronic prescribing system in primary care: A qualitative assessment”, for publication in BMC Medical Informatics and Decision-Making.

Please let us know if we can provide any further modifications.

Thank you kindly.

Sincerely yours,

Emily Beth Devine

EB Devine, PharmD, MBA, PhD  
Associate Professor  
Phone: 1-206-221-5760  
Fax:1-206-543-3835  
bdevine@uw.edu